Author's response to reviews

Title: Dietary preference, physical activity, and cancer risk in men: national health insurance corporation study

Authors:

Young Ho Yun (lawyun08@ncc.re.kr)
Min Kyung Lim (mickey@ncc.re.kr)
Young-Joo Won (astra67@ncc.re.kr)
Sang Min Park (sangmin.park.snuh@gmail.com)
Yoon Jung Chang (eunicemd@ncc.re.kr)
Sang Woo Oh (osw6021@gmail.com)
Soon Ae Shin (rosa1026@nhic.or.kr)

Version: 7 Date: 17 November 2008

Author's response to reviews:

November 17, 2008

Melissa Norton, MD
Editor-in-Chief, BMC Cancer
BioMed Central Ltd, Middlesex House,
34-42 Cleveland Street, London W1T 4LB, UK.

Dear Dr Melissa Norton:

Thank you for accepting our paper.
We just changed several colors of lines in Table 2-5.
We look forward to having our manuscript published in BMC Cancer.

Sincerely yours,

Young Ho Yun, MD, PhD
Division of Cancer Control
National Cancer Control Research Institute
National Cancer Center
809, Madu-dong, Ilsan-gu, Goyang-si
Gyeonggi-do, 411-769, Korea
Tel: +82-31-920-1705 Fax: +82-31-920-2199
E-mail address: lawyun08@ncc.re.kr